Author's response to reviews

Title: Long-term health-related and economic consequences of short-term outcomes in evaluation of perinatal interventions.

Authors:

Margreet J. Teune (m.j.teune@amc.uva.nl)
Aleid G. van Wassenaer (a.vanwassenaer@amc.uva.nl)
Ben-Willem J. Mol (b.w.mol@amc.uva.nl)
Brent C. Opmeer (b.c.opmeer@amc.uva.nl)

Version: 6 Date: 23 July 2010

Author's response to reviews: see over
Editor-in-Chief
BMC Pregnancy and Childbirth

Amsterdam, 23 July 2010

Subject: Study Protocol “Long-term health-related and economic consequences of short-term outcomes in evaluation of perinatal interventions”.
ZonMW projectnumber: 80-82325-98-9010.

Dear Miss Titmus,

Thank you for accepting our manuscript for publication.
The following changes we have made in the format of our manuscript:

1. We indicate that all the authors have read and approved the final manuscript.
2. We have cited our references in squared brackets.
3. We deleted the line numbers.
4. We have removed the additional copy of our funding.
5. We removed all highlighting from our manuscript.
6. We removed the term et al. in our reference list.

We hope that our manuscript is now ready for publication. Do not hesitate to contact me if you have any further questions on our manuscript.

With best wishes,

Margreet Teune (corresponding author)

M.J. Teune, MD, MSc
Academic Medical Centre
Dept Obstetrics and Gynaecology, Room H4-140
Meibergdreef 9
1105 AZ Amsterdam
The Netherlands
E: M.J.Teune@amc.uva.nl, margreet@teune.nl
T: +31-(0)20-566199
M: +31-(0)6-47134234